

*Citation for published version:* Chantratita, N, Wuthiekanun, V, Limmathurotsakul, D, Vesaratchavest, M, Thanwisai, A, Amornchai, P, Tumapa, S, Feil, EJ, Day, NP & Peacock, SJ 2008, 'Genetic Diversity and Microevolution of Burkholderia pseudomallei in the Environment', *PLoS Neglected Tropical Diseases*, vol. 2, no. 2, pp. e182. https://doi.org/10.1371/journal.pntd.0000182

DOI: 10.1371/journal.pntd.0000182

Publication date: 2008

Link to publication

```
Publisher Rights
CC BY
```

Citation: Chantratita N, Wuthiekanun V, Limmathurotsakul D, Vesaratchavest M, Thanwisai A, et al. (2008) Genetic Diversity and Microevolution of Burkholderia pseudomallei in the Environment. PLoS Negl Trop Dis 2(2): e182. doi:10.1371/journal.pntd.0000182

© 2008 Chantratita et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**University of Bath** 

# Alternative formats

If you require this document in an alternative format, please contact: openaccess@bath.ac.uk

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

#### Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

# Genetic Diversity and Microevolution of *Burkholderia pseudomallei* in the Environment

Narisara Chantratita<sup>19</sup>, Vanaporn Wuthiekanun<sup>19</sup>, Direk Limmathurotsakul<sup>1</sup>, Mongkol Vesaratchavest<sup>1</sup>, Aunchalee Thanwisai<sup>1</sup>, Premjit Amornchai<sup>1</sup>, Sarinna Tumapa<sup>1</sup>, Edward J. Feil<sup>2</sup>, Nicholas P. Day<sup>1,3</sup>, Sharon J. Peacock<sup>1,3</sup>\*

1 Mahidol-Oxford Tropical Medicine Research Unit, Faculty of Tropical Medicine, Mahidol University, Bangkok, Thailand, 2 Department of Biology and Biochemistry, University of Bath, Bath, United Kingdom, 3 Center for Clinical Vaccinology and Tropical Medicine, Nuffield Department of Clinical Medicine, University of Oxford, Churchill Hospital, Oxford, United Kingdom

# Abstract

**Background:** The soil dwelling Gram-negative pathogen *Burkholderia pseudomallei* is the cause of melioidosis. The diversity and population structure of this organism in the environment is poorly defined.

*Methods and Findings:* We undertook a study of *B. pseudomallei* in soil sampled from 100 equally spaced points within 237.5 m<sup>2</sup> of disused land in northeast Thailand. *B. pseudomallei* was present on direct culture of 77/100 sampling points. Genotyping of 200 primary plate colonies from three independent sampling points was performed using a combination of pulsed field gel electrophoresis (PFGE) and multilocus sequence typing (MLST). Twelve PFGE types and nine sequence types (STs) were identified, the majority of which were present at only a single sampling point. Two sampling points contained four STs and the third point contained three STs. Although the distance between the three sampling points was low (7.6, 7.9, and 13.3 meters, respectively), only two STs were present in more than one sampling point. Each of the three samples was characterized by the localized expansion of a single *B. pseudomallei* clone (corresponding to STs 185, 163, and 93). Comparison of PFGE and MLST results demonstrated that two STs contained strains with variable PFGE banding pattern types, indicating geographic structuring even within a single MLST-defined clone.

*Conclusions:* We discuss the implications of this extreme structuring of genotype and genotypic frequency in terms of micro-evolutionary dynamics and ecology, and how our results may inform future sampling strategies.

Citation: Chantratita N, Wuthiekanun V, Limmathurotsakul D, Vesaratchavest M, Thanwisai A, et al. (2008) Genetic Diversity and Microevolution of *Burkholderia* pseudomallei in the Environment. PLoS Negl Trop Dis 2(2): e182. doi:10.1371/journal.pntd.0000182

Editor: Bart Currie, Charles Darwin University, Australia

Received June 13, 2007; Accepted January 7, 2008; Published February 20, 2008

**Copyright:** © 2008 Chantratita et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This study was funded by The Wellcome Trust of Great Britain. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\*E-mail: sharon@tropmedres.ac

<sup>®</sup>These authors contributed equally to this work.

# Introduction

The soil dwelling Gram-negative bacterium Burkholderia pseudo*mallei* is the cause of melioidosis. This organism is present in the environment across much of southeast Asia and Northern Australia and is increasingly recognised elsewhere, including parts of South America [1,2]. Infection occurs through bacterial inoculation and contamination of wounds, and more rarely by inhalation and ingestion [1,3]. Environmental sampling underpins efforts to define the global distribution of B. pseudomallei in soil, and the associated geographic distribution of risk to humans and livestock. Sampling is also performed during the investigation of suspected outbreaks, when bacterial genotyping is used to compare B. pseudomallei obtained from cases of melioidosis with strains from a specified environment or substance. Environmental sampling would also be required following the deliberate release of B. pseudomallei associated with bioterrorist activity. The accuracy of such studies depends on the detection of all of the B. pseudomallei genotypes present at a given site with the exception of those

present at an extremely low frequency. Informed sampling strategies are also a prerequisite for meaningful comparisons between environmental isolates and those recovered from cases of disease in humans and animals, which provide an important means to identify clones with heightened virulence.

The objective of most published environmental studies of *B. pseudomallei* has been to determine its presence based on culture of soil and/or water in different geographic regions, particularly in Southeast Asia and northern Australia [4–17]. Several studies were conducted prior to the initial recognition [18], and later description in 1998 of *B. thailandensis* [19], a putatively non-virulent but closely related species present in the soil which can cause confusion because it has very similar colony morphology characteristics to *B. pseudomallei* on solid media. Yield of *B. pseudomallei* from different soil depths and during different times of the year have also been examined [8,12], and quantitative culture of *B. pseudomallei* has been performed in several countries [9,14,16,17]. The combined results of these studies indicate that *B. pseudomallei* count in soil varies with sampling depth and

## **Author Summary**

The soil dwelling Gram-negative bacterium Burkholderia pseudomallei is the cause of melioidosis, a serious human infection that occurs in Southeast Asia and northern Australia. The purpose of this study was to evaluate the population genetic structure of *B. pseudomallei* in the environment. To achieve this, we undertook soil sampling and culture for the presence of B. pseudomallei in 100 equally spaced points within an area of disused land in northeast Thailand, and undertook detailed genotyping of primary plate colonies isolated from three independent sampling points. Our results demonstrated that multiple B. pseudomallei genotypes were present within a single soil sample, and that different genotypes were present at independent but nearby sampling points. The B. pseudomallei genetic population was unevenly distributed within a given sample, with a predominant genotype co-existing with several genotypes present as a minority population. We discuss the implications of this structuring of genotypic frequency in terms of micro-evolutionary dynamics and ecology, and how our results may inform future sampling strategies.

calendar period (and associated weather conditions), and varies in presence and quantity both within and between countries. Environmental sampling has also been employed during an investigation of a suspected outbreak of melioidosis in northern Australia in which a case cluster was linked to the water supply through genotyping by pulsed field gel electrophoresis (PFGE) [20], and as part of an environmental sampling program in the same region [21]. Two studies have also compared bacterial genotypes of environmental versus disease-associated strains using ribotyping or multilocus sequence typing (MLST) [22,23]. The soil sampling methodology used previously to detect and in some cases quantify B. pseudomallei using culture methods has varied. The quantity of soil sampled ranged from 3 g to 100 g, although the addition of water to the sample, use of the supernatant for culture, and the ratio of water to soil used are common to many studies. An exception was the identification of B. pseudomallei following passage through hamsters inoculated with soil extracts [15].

Despite the importance of environmental sampling for the presence of *B. pseudomallei*, the genetic variability of this organism within a single sample or between adjacent sampling points is not known, and the strategies necessary to ensure sampling of a genetically unbiased *B. pseudomallei* population are undefined. The aim of this study conducted in northeast Thailand was to address these issues. We describe the presence of multiple *B. pseudomallei* genotypes within a single soil sample, and the presence of different *B. pseudomallei* genetypes at independent but nearby sampling points. The *B. pseudomallei* genetic population was unevenly distributed within a given sample, with a predominant genotype co-existing with several genotypes present as a minority population. We discuss the implications of this structuring of genotypic frequency in terms of micro-evolutionary dynamics and ecology, and how our results may inform future sampling strategies.

## Methods

#### Study site

Soil samples were collected during September 2005 (the rainy season) from an area of disused land measuring 237.5 m<sup>2</sup> (23.75 m×10 m) situated to one side of road 231 in Amphoe Meung, at a distance of 8 km northeast of Sappasithiprasong Hospital, Ubon Ratchathani, northeast Thailand. This is a rural

rice-growing region where road traffic is light and cattle range through the area. The soil type was sandy loam, and was wet under foot but not flooded. The vegetation was low-lying scrub and the area showed no signs of cultivation. The site included a single concrete electricity pole. A brick wall formed one boundary, running parallel to and distal from the road.

Five people undertook sampling over a 2-day period of intermittent rain. The site was initially divided into a grid system using string and wooden stakes, in which  $5 \times 20$  spots were plotted 2.5 m apart on the vertical axis, and 1.25 m apart on the horizontal axis. The grid was referenced alphabetically (A to E for horizontal rows as viewed back against the wall, row E lying closest to road) and numerically (1 to 20, moving across from left to right on vertical axis). Each point is hereafter termed a 'sampling point' and the specific site defined by its grid reference.

#### Soil sampling

A hole was dug with a clean spade to a depth of approximately 30 centimetres. A clean plastic bag was placed on weighing scales and a sample of soil (100 grams) was removed from the base of the hole and placed into the bag. Each soil sample was labelled using pre-prepared stickers denoting the grid reference number. The bag was closed and stored out of direct sunlight at ambient temperature until transported to the laboratory where it was processed on the same day. The utensils used for sampling were cleaned between each use by rinsing with bottled water to remove visible debris, followed by cleaning with 70% ethanol and air drying.

#### Soil culture and B. pseudomallei identification

Soil samples were batch processed at the end of each collection day. 100 ml of sterile water was added to each bag, mixed well and left overnight to sediment. The upper layer of water was then transferred by plastic pipette to a sterile plastic container. Four aliquots of 100 µl were spread plated onto each of 4 Ashdown's selective agar plates. A further 1 ml of the soil water sample was added to 9 ml of selective enrichment broth consisting of threoninebasal salt plus colistin (TBSS-C50 broth). This was incubated at  $40^{\circ}$ C in air for 48 h, after which 10 µl of surface liquid was plated onto a second Ashdown's agar plate which was incubated and observed as before. Agar plates were incubated at 40°C in air and visually inspected daily for 4 days. Colonies of B. pseudomallei were initially identified on the basis of colony morphotype. This included the characteristic colony morphology (purple, flat, dry and wrinkled) together with 6 additional colony morphotypes, as described previously [24]. Colonies suspected to be B. pseudomallei were tested using the oxidase test, and positive colonies confirmed as B. pseudomallei using a highly specific latex agglutination test (positive for B. pseudomallei but negative for B. thailandensis) [25,26].

#### Genotyping of B. pseudomallei

Genotyping of *B. pseudomallei* was performed for 3 sampling points (grid reference A11, D10 and E4). These were selected at random from sampling points that gave at least 200 *B. pseudomallei* colonies on the two primary Ashdown's agar plates. For each sample, 200 primary colonies were picked to purity and subjected to PFGE using *SpeI*, as previously described [27]. Analysis of PFGE banding patterns for the 200 colonies at each of the three sampling points was performed using the BioNumerics software version 2.5 (Applied Maths, Belgium). For the purposes of this study, interpretation was defined so as to be highly discriminatory. Isolates with identical PFGE banding patterns were regarded as genotypically indistinguishable, but isolates with one or more bands different were defined as putatively different and given a different banding pattern number. One bacterial representative of each banding pattern type was further



Figure 1. The presence of *B. pseudomallei* in 100 spaced sampling points within an area of disused land in northeast Thailand. doi:10.1371/journal.pntd.0000182.g001

examined using MLST, as previously described [23]. The alleles at each of the loci were assigned and sequence type defined using the *B. pseudomallei* MLST website (http://bpseudomallei.mlst.net).

# Measures of genetic diversity

Genetic diversity of B. pseudomallei within a given sampling point was defined using Simpson's index of diversity. This describes the probability that two randomly selected bacterial cells within a sampling point will be different genotypes; 0 indicates no diversity (all cells identical) and 1 indicates maximum diversity (all cells different). Confidence intervals for Simpson's index were calculated as described previously [28]. Further analysis was performed to examine whether the genetic distance between isolates within a given sample was significantly different from that expected if all isolates were randomly distributed between the three sites. The number of different alleles was determined for all 19,900 pairwise comparisons of the 200 colonies (strains) at each of the three sampling points. The results were compared to mean values calculated from 100 random samples, each of 200 strains, drawn with replacement from the combined data set of 600 strains (all 3 sampling sites). Statistical significance was gauged by calculating the 0.01, 0.05 and 0.95 and 0.99 percentiles from the re-sampled data. Genetic diversity between two sampling points was measured using the Morisita index of similarity. This ranges from 0 to 1; 0 indicates that no genotypes are shared between the two sampling points, and 1 indicates complete identity. All analyses were carried out using Stata 9.0 (College Station, Texas, United States), except the random resampling procedure which used a PERL script written by EJF (available on request).

#### Results

A total of 80 out of the 100 sampling points were culture positive for *B. pseudomallei*, of which 77 were positive from both direct plating onto Ashdown's agar and selective enrichment broth, and 3 were positive from selective enrichment broth culture alone (Figure 1). *B. thailandensis* was not detected. The genetic variability of *B. pseudomallei* was defined and compared within and between sampling points by genotyping 200 colonies at each of three positive points (A11, D10 and E4, see Figure 1). PFGE of 600 individual primary colonies revealed 12 PFGE banding pattern types. MLST of a single random isolate of each of the 12 PFGE types revealed 9 distinct sequence types (STs) (Table 1).

Table 2 shows the breakdown of STs within each of the three sampling points. D10 and E4 each contained four STs and A11 contained three STs. Although the distance between the 3 sampling points was low (7.6, 7.9 and 13.3 meters for A11-D10, D10-E4, and A11-E4, respectively), only two STs were present in more than one sampling point (E4/D10; ST176, and D10/A11; ST60); no STs were detected in all three points, and no STs were common to E4 and A11 which were the two sites separated by the greatest distance. This strong segregation of STs was reflected in low Morisita index values (Table 2). Furthermore, each site was characterized by the following predominant genotypes, each of which was restricted to a single site: ST93 in A11 (87%), ST163 in D10 (51.5%), and ST185 in E4 (70%). Simpson's index of diversity ranged from 0.24–0.65 (Table 2).

The finding of very limited overlap between the sampling points was further examined by comparing the average pairwise distance

PFGE Type	Sequence Type	MLST Profile						
		ace	gltB	gmhD	lepA	lipA	narK	ndh
1	ST 424	4	12	10	4	8	3	1
2	ST 177	1	1	4	3	1	3	1
3	ST 176	3	1	4	1	1	3	1
4	ST 185	1	4	2	2	1	4	1
5	ST 33	1	4	12	1	1	2	1
6, 11, 12	ST 60	3	1	12	1	1	3	1
7	ST 163	3	2	2	1	1	4	1
8, 10	ST 93	1	1	2	1	1	4	1
9	ST 304	1	1	5	1	1	4	1

**Table 1.** Relationship between PFGE and MLST analysis of soil isolates.

doi:10.1371/journal.pntd.0000182.t001

**Table 2.** Genotyping results for 200 colonies of *B. pseudomallei* from each of three independent sampling points.

Sequence Type	Sampling Points					
	E4	D10	A11			
ST 424	38 (19%)					
ST 177	12 (6%)					
ST 176	10 (5%)	29 (14.5%)				
ST 185	140 (70%)					
ST 33		50 (25%)				
ST 60		18 (9%)	9 (4.5%)			
ST 163		103 (51.5%)				
ST 93			174 (87%)			
ST 304			17 (8.5%)			
Simpson Index of diversity (95% Cl)	0.47 (0.40–0.54)	0.65 (0.60–0.69)	0.24 (0.16-0.31)			
Morisita Index of sim	ilarity					
- compared to E4	-	0.02	0.00			
- compared to D10	-	-	0.01			

doi:10.1371/journal.pntd.0000182.t002

(in terms of allelic mismatches) between isolates within each point to that expected if the three points were combined as a single population. Figure 2 shows the proportion of all 19,900 ((200\*199)/2) pairwise comparisons showing 0, 1, 2 ... 7 allelic mismatches for each of the seven MLST loci for 200 colonies (strains) examined at each of three independent sampling points. This was compared to randomized data derived from the mean



values for 100 random samples of 200 strains resampled with replacement from the combined data set of 600 strains from all three points. All three sampling points showed a significantly greater proportion of identical pairs (i.e. the same ST) than the randomized data (P<0.01), which is consistent with localised clonal expansion. For two of the three sampling sites, no significant difference was observed in the proportion of related but nonidentical STs (i.e. those differing at a single locus) compared with the randomized data. The exception was A11 where a significantly higher number of pairwise comparisons (P<0.01) corresponded to a single MLST locus difference; ST304 is a single locus variant of the predominant clone ST93, being variant at gmhD. Comparison of the two gmhD alleles (allele 2 for ST93 and allele 5 for ST304) indicated two base differences (C $\rightarrow$ T position 118, and T $\rightarrow$ C position 327). The third genotype noted at A11 is ST60; this differs at three loci from both ST93 and ST304, thus accounting for the small peak at three mismatches in the A11 plot.

Comparison of PFGE and MLST results demonstrated that two STs contained strains with variable PFGE banding pattern types. ST93 contained strains with two PFGE types (types 8 and 10), which were 12 bands different. This ST was only found in sampling point A11, in which the proportion of each banding pattern was 172/174 (99%) for type 8, and 2/174 (1%) for type 10. ST60 contained strains with three PFGE types (types 6, 11 and 12). The difference in banding patterns between these three was 6 bands (PFGE type 6 versus 11), 1 band (PFGE type 6 versus 12), and 2 bands (PFGE type 11 versus 12). All 18 ST60 isolates from D10 corresponded to type 6, whereas 8/9 ST60 isolates from A11 corresponded to type 11, and 1/9 to type 12. These PFGE data confirm fine-scale geographic structuring, even within a single MLST-defined clone. We also note differences between the single locus variants ST 304 (type 9) and ST93 (types 8 and 10); PFGE type 9 differed from type 8 by 18 bands, and from type 10 by 10 hands



**Figure 2. Graph of the proportion of all pairwise comparisons showing 0, 1, 2 ... 7 allelic mismatches for each of 200 primary colonies (strains) examined at three independent sampling points.** The "trials" data represents mean values for 100 random samples of 200 strains drawn from the combined data set of 600 strains from all three sampling points (with replacement). Error bars are based on the 5<sup>th</sup> and 95<sup>th</sup> percentiles of the 100 random samples. No pw comparisons (real or trial data) differ at all seven loci since the locus *ndh* is monomorphic (invariant). doi:10.1371/journal.pntd.0000182.g002

#### Discussion

This study has demonstrated marked geographic structuring of B. pseudomallei genotypes in soil. The dramatic differences in genotype frequency over such small distances are striking, but difficult to interpret. One explanation is that the numerically dominant ST at each sampling point represents a strain with superior biological fitness compared with STs present as a minority of the population. This could relate to factors such as soil type or pH, or competition with other microbial species. This would assume that adjacent foci of soil have variable microenvironments, but it seems unlikely that nearby sampling points within a confined area of disused land would differ sufficiently to support multiple, non-overlapping niches. An alternative possibility is that of local competition between clones of B. pseudomallei. Flooding or other disturbance mechanisms would provide the means for a given clone to migrate and become established within a specific plot. Once the clone has reached a certain threshold frequency, it could repel invaders either by the production of microbicides, through phage to which they themselves are resistant, or via other killing mechanisms. The presence of a clone as a minority population could represent the ability of this strain to survive at a lower level, or could represent the boundary of a point of predominance in an adjacent point or focus.

This study also provides evidence for microevolution of B. *pseudomallei* in soil. PFGE is a more sensitive marker of very rapid genetic change than MLST in this species. Alterations in banding pattern arise due to any kind of genetic event that alters the presence or absence of restriction sites anywhere in the genome, or else changes the distance between existing sites. In contrast, MLST genes are chosen specifically to code for a central housekeeping role and to be highly conserved. MLST is therefore blind to largescale genomic rearrangements that may dramatically alter the PFGE banding pattern [29]. Two of the nine STs contained strains with variable PFGE banding pattern types. We postulate that these changes represent microevolution within our sampling site rather than importation of several strains with matching ST but a different banding pattern. This is consistent with the finding that genomic islands constitute  $\sim 6\%$  of the *B. pseudomallei* K96243 genome [30]. Furthermore, comparison of the whole genome sequences of B. pseudomallei and B. mallei indicated the capacity for genomic rearrangement and gene loss by two species that are highly related by MLST [31]. Our findings are consistent with a dynamic genome that is evolving through the movement of genomic islands and rearrangements such as inversions and indels.

The co-existence in a single soil sample of a single locus variant of ST93 (ST304) can be explained by *in situ* microevolution or by a chance association. ST93 and ST304 have both been isolated previously in northeast Thailand. We recovered ST93 from the environment in 1990, 1998 and 1999, each from different sampling sites situated along road 212 which runs northwest from the town of Ubon Ratchathani. The MLST database (www.mlst. net) contains a fourth ST93 isolate that was associated with human disease in Thailand in 1998. We have also recovered ST304 from two patients with melioidosis presenting to a hospital in northeast Thailand in 1999. However, an accurate picture of the distribution and frequency of co-localization in soil of ST93 and ST304 in this region has not been defined, and it is difficult to speculate on the probability of a chance association.

A potential pitfall of this study is that the proportion of each ST was obtained after the soil sample had been prepared by mixing with water and overnight sedimentation followed by growth using

#### References

rich media. Some STs may be more adapted to survival or growth after the addition of distilled water during sample preparation, or may move more efficiently into the layer of surface water that is removed for culture. It is also possible that some STs are more likely to grow on laboratory media than others, and that some STs are viable but non-culturable under the conditions used. Resolution of these issues will require the direct application of molecular tools to soil samples, and comparison of genotypes with those obtained using conventional culture and existing soil preparation methods.

Our findings have several important implications for future genotyping studies. Soil sampling at a single location will fail to identify the genotypes present at a distance of even a few meters. Furthermore, the predominance of a single ST at a given site requires that extra sampling effort is required to detect any genotypes present as a minority of the population. We estimate that the characterization of approximately 50 colonies from any single site would provide an 85% probability of detecting a genotype present at the site at a frequency of 2%. This is based on the exact 95% binomial confidence interval for ST60 at sampling point A11, which was present in the lowest proportion. The temporal stability of the genotype distribution described here is not known, and it is possible that markedly different genotypic frequencies might be recorded from the same sampling sites if the study were to be repeated at some point in the future. It is also unclear whether the degree of genetic diversity described here will be reproduced within Thailand and in other endemic countries, although a study by Pearson et al. in which genetic diversity was demonstrated by variable number tandem repeat (VNTR) analysis within a small geographic area of Australia [32] suggests that this will prove to be a reproducible finding. The basis on which PCR is used in future studies to detect *B. pseudomallei* in the environment also requires careful consideration. Amplification may give rise to mixed products, and DNA from strains present at low copy number may go undetected. PCR may become an appropriate technology for the detection of B. pseudomallei, but is not an appropriate basis for subsequent genotyping unless multiple independent amplicons are evaluated.

This study has investigated *B. pseudonallei* in soil taken from an area of disused land. This is in contrast to many previous studies in Thailand which were conducted in rice paddies. The basis for our choice was to examine an environment free of external influences such as chemical fertilizers and pesticides, together with the effect of ploughing, planting, burning of rice stubble and the presence of rice plants. However, most disease is probably acquired in rice paddies; further studies are underway to compare and contrast the findings reported here with those from a rice paddy in the same region.

# Acknowledgments

We are grateful to staff at the Mahidol-Oxford Tropical Medicine Research Unit for their support. We thank Wirichada Pan-ngum for statistical advice.

# **Author Contributions**

Conceived and designed the experiments: ND NC VW MV DL EF SP. Performed the experiments: NC VW MV DL AT PA ST. Analyzed the data: ND NC VW MV DL EF AT PA ST SP. Contributed reagents/ materials/analysis tools: NC VW DL EF. Wrote the paper: ND SP NC VW DL EF.

 Cheng AC, Currie BJ (2005) Melioidosis: epidemiology, pathophysiology, and management. Clin Microbiol Rev 18: 383–416.

Rolim DB, Vilar DC, Sousa AQ, Miralles IS, de Oliveira DC, et al. (2005) Melioidosis, northeastern Brazil. Emerg Infect Dis 11: 1458–1460.

- Wiersinga WJ, van der Poll T, White NJ, Day NP, Peacock SJ (2006) Melioidosis: insights into the pathogenicity of *Burkholderia pseudomallei*. Nat Rev Microbiol 4: 272–282.
- Strauss JM, Groves MG, Mariappan M, Ellison DW (1969) Melioidosis in Malaysia. II. Distribution of *Pseudomonas pseudomallei* in soil and surface water. Am J Trop Med Hyg 18: 698–702.
- Thomas AD (1977) The isolation of *Pseudomonas pseudomallei* from soil in North Queensland. Aust Vet J 53: 408.
- Nachiangmai N, Patamasucon P, Tipayamonthein B, Kongpon A, Nakaviroj S (1985) *Pseudomonas pseudomallei* in southern Thailand. Southeast Asian J Trop Med Public Health 16: 83–87.
- Merianos A, Patel M, Lane JM, Noonan CN, Sharrock D, et al. (1993) The 1990-1991 outbreak of melioidosis in the Northern Territory of Australia: epidemiology and environmental studies. Southeast Asian J Trop Med Public Health 24: 425–435.
- Wuthiekanun V, Smith MD, Dance DAB, White NJ (1995) Isolation of *Pseudomonas pseudomallei* from soil in northeastern Thailand. Trans R Soc Trop Med Hyg 89: 41–43.
- Smith MD, Wuthiekanun V, Walsh AL, White NJ (1995) Quantitative recovery of Burkholderia pseudomallei from soil in Thailand. Trans R Soc Trop Med Hyg 89: 488–490.
- Yang S, Tong S, Lu Z (1995) Geographical distribution of *Pseudomonas* pseudomallei in China. Southeast Asian J Trop Med Public Health 26: 636–638.
- Vuddhakul V, Tharavichitkul P, Na-Ngam N, Jitsurong S, Kunthawa B, et al. (1999) Epidemiology of *Burkholderia pseudomallei* in Thailand. Am J Trop Med Hyg 60: 458–461.
- Brook MD, Currie B, Desmarchelier PM (1997) Isolation and identification of Burkholderia pseudomallei from soil using selective culture techniques and the polymerase chain reaction. J Appl Microbiol 82: 589–596.
- Heng BH, Goh KT, Yap EH, Loh H, Yeo M (1998) Epidemiological surveillance of melioidosis in Singapore. Ann Acad Med Singapore 27: 478–484.
- Parry CM, Wuthiekanun V, Hoa NT, Diep TS, Thao LT, et al. (1999) Melioidosis in Southern Vietnam: clinical surveillance and environmental sampling. Clin Infect Dis 29: 1323–1326.
- Finkelstein RA, Atthasampunna P, Chulasamaya M (2000) Pseudomonas (Burkholderia) pseudomallei in Thailand, 1964–1967: geographic distribution of the organism, attempts to identify cases of active infection, and presence of antibody in representative sera. Am J Trop Med Hyg 62: 232–239.
- Wuthiekanun V, Mayxay M, Chierakul W, Phetsouvanh R, Cheng AC, et al. (2005) Detection of *Burkholderia pseudomallei* in soil within the Lao People's Democratic Republic. J Clin Microbiol 43: 923–924.
- Chierakul W, Winothai W, Wattanawaitunechai C, Wuthiekanun V, Rugtaengan T, et al. (2005) Melioidosis in 6 tsunami survivors in southern Thailand. Clin Infect Dis 41: 982–990.

- Smith MD, Angus BJ, Wuthiekanun V, White NJ (1997) Arabinose assimilation defines a nonvirulent biotype of *Burkholderia pseudomallei*. Infect Immun 65: 4319–4321.
- Brett PJ, DeShazer D, Woods DE (1998) Burkholderia thailandensis sp. nov., a Burkholderia pseudomallei-like species. Int J Syst Bacteriol 48: 317–320.
- Currie BJ, Mayo M, Anstey NM, Donohoe P, Haase A, et al. (2001) A cluster of melioidosis cases from an endemic region is clonal and is linked to the water supply using molecular typing of *Burkholderia pseudomallei* isolates. Am J Trop Med Hyg 65: 177–179.
- Inglis TJ, Foster NF, Gal D, Powell K, Mayo M, et al. (2004) Preliminary report on the northern Australian melioidosis environmental surveillance project. Epidemiol Infect 132: 813–820.
- Trakulsomboon S, Dance DA, Smith MD, White NJ, Pitt TL (1997) Ribotype differences between clinical and environmental isolates of *Burkholderia pseudomallei*. J Med Microbiol 46: 565–570.
- Vesaratchavest M, Tumapa S, Day NP, Wuthickanun V, Chierakul W, et al. (2006) Nonrandom distribution of *Burkholderia pseudomallei* clones in relation to geographical location and virulence. J Clin Micro 44: 2553–2557.
- geographical location and virulence. J Clin Micro 44: 253-2557.
   Chantratita N, Wuthiekanun V, Boonbumrung K, Tiyawisutsri R, Vesaratchavest M, et al. (2007) Biological relevance of colony morphology and phenotypic switching by *Burkholderia pseudomallei*. J Bacteriol 189: 807-812.
- Wuthiekanun V, Anuntagool N, White NJ, Sirisinha S (2002) Short report: a rapid method for the differentiation of *Burkholderia pseudomallei* and *Burkholderia* thailandensis. Am J Trop Med Hyg 66: 759–761.
- Anuntagool N, Naigowit P, Petkanchanapong V, Aramsri P, Panichakul T, et al. (2000) Monoclonal antibody-based rapid identification of *Burkholderia pseudomallei* in blood culture fluid from patients with community-acquired septicaemia. J Med Microbiol 49: 1075–1078.
- Maharjan B, Chantratita N, Vesaratchavest M, Cheng A, Wuthiekanun V, et al. (2005) Recurrent meliodosis in patients in northeast Thailand is frequently due to reinfection rather than relapse. J Clin Microbiol 43: 6032–6034.
- Grundmann H, Hori S, Tanner G (2001) Determining confidence intervals when measuring genetic diversity and the discriminatory abilities of typing methods for microorganisms. J Clin Microbiol 39: 4190–4192.
- Cooper JE, Feil EJ (2004) Multilocus sequence typing–what is resolved? Trends Microbiol 12: 373–377.
- Holden MT, Titball RW, Peacock SJ, Cerdeno-Tarraga AM, Atkins T, et al. (2004) Genomic plasticity of the causative agent of melioidosis, *Burkholderia pseudomallei*. Proc Natl Acad Sci USA 101: 14240–14245.
- Nierman WC, DeShazer D, Kim HS, Tettelin H, Nelson KE, et al. (2004) Structural flexibility in the *Burkholderia mallei* genome. Proc Natl Acad Sci USA 101: 14246–14251.
- Pearson T, U'Ren JM, Schupp JM, Allan GJ, Foster PG, et al. (2007) VNTR analysis of selected outbreaks of *Burkholderia pseudomallei* in Australia. Infection, Genetics and Evolution 7: 416–423.